Author’s response to reviews

Title: Pretreatment but not subsequent coincubation with midazolam reduces the cytotoxicity of temozolomide in neuroblastoma cells.

Authors:

Sebastian Braun (brauns@uni-duesseldorf.de)

Inge Bauer (Inge.Bauer@med.uni-duesseldorf.de)

Benedikt Pannen (Benedikt.Pannen@med.uni-duesseldorf.de)

Robert Werdehausen (Robert.Werdehausen@med.uni-duesseldorf.de)

Version: 2 Date: 29 Sep 2015

Author’s response to reviews:

Dear Dr. Lopes de Sousa,

we are very grateful for your reply. Please find the revised manuscript with the modified title page. The modification is highlighted and was adapted to the instructions for authors (research articles) of BMC Anesthesiology. We hope that we fulfill this requirement and look forward to your answer.

Sincerely,

Dr. Sebastian Braun